Author’s response to reviews

Title: The medical diagnosis as a linguistic game

Authors:

Peter Fritz (peter.fritz23@arcor.de)
Andreas Kleinhans (andreas.kleinhans@fuchspartner.de)
Florian Kuisle (florian.kuisle@gmx.de)
Christine Fritz-Kuisle (christine-fritz@web.de)
Mark Dominik Alscher (dominik.alscher@rbk.de)

Version: 5 Date: 11 Jun 2017

Author’s response to reviews:

MIDM-D-16-00235R4

The medical diagnosis as a linguistic game

Peter Fritz, MD; Andreas Kleinhans, MD; Florian Kuisle, MD; Patrick Albu, Dipl.-Infor.; Christine Fritz-Kuisle, MD; Mark Dominik Alscher, MD

BMC Medical Informatics and Decision Making

Dear Editor,

you raised the following questions. I answer point by point:

1. Author contributions

Thank you for providing this, however, we require that the authors each be referred to by their initials- you should ensure that the initials of no two of the authors are the same to ensure their identification.

Answer: Corrected.
2. Consent to publish

Thank you for including this sub-section of the Declarations, however, this does not refer to the consent that authors give to publish their manuscript. This actually refers to the consent that participants give for the publication of their identifying data - as you have not used such data from patients, you may state 'Not applicable' in this section.

Answer: Corrected.

3. Ethics approval and consent to participate

Please ensure that you include the full names and affiliations of all the ethics committees that approved the study and the committees' reference number if appropriate. If you did not need formal ethics approval please confirm that this complies with national guidelines and provide a reference which supports this. Alternatively, supply a statement that says that a local ethics committee ruled that no formal ethics approval was required in this particular case.

Answer: Corrected.

4. Final manuscript file

On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

Answer: Corrected.

Kind regards

Dominik